Author's response to reviews

Title: Strategies to reduce clinical inertia in hypertensive kidney transplant recipients

Authors:

   James Kiberd (bryce.kiberd@dal.ca)
   Romauld Panek (romauld.panek@cdha.ca)
   Bryce Kiberd (bryce.kiberd@dal.ca)

Version: 4 Date: 24 July 2007

Author's response to reviews:

Changes made

1. Title now has sentence case.
2. Running title deleted.
3. Acknowledgements added (None); Funding source added (None)
4. Reference 12 space added between Levey AS.
5. Table title and text now have sentence case.
6. Hyphen inserted between abbreviations and full names.